Author's response to reviews

Title: Primary mucinous adenocarcinoma of the thymic gland with intestinal differentiation and HLA-DRB5-locus deletion. An immunohistochemical and molecular study with review of the Literature

Authors:

Maryam Maghbool (maghbol.maryam@yahoo.com)
Mani Ramzi (ramzim@sums.ac.ir)
inga Nagel (inagel@medgen.uni-kiel.de)
Pablo Bejarano (pbejaran@med.miami.edu)
Reiner Siebert (rsiebert@medgen.uni-kiel.de)
Yahya Daneshbod (y@daneshbod.com)
Abolfazl Saeedzadeh (asaeedzadeh@gmail.com)

Version: 2 Date: 8 January 2013

Author's response to reviews:

Dear Chief Editor

Please kindly find my revised article entitled: "Primary mucinous adenocarcinoma of the thymic gland with intestinal differentiation and HLA-DRB5-locus deletion. An immunohistochemical and molecular study with review of the Literature". The following changes have been made according to editorial:

1-consent statement
2-IDENTIFYING INFORMATION in figure B is removed
3-Competing interests section is added.
4-Authors' contributions section is added
5-Acknowledgements section is added.

Thank you very much.

Y. Daneshbod MD